Author's response to reviews

Title: Desmoid tumors complicating Familial Adenomatous Polyposis: a meta-analysis of the mutation spectrum of at-risk individuals

Authors:

Voytek Slowik (vslowik@cmh.edu)
Thomas Attard (tmattard@cmh.edu)
Dai Hongying (hdai@cmh.edu)
Raj Shah (shahra@umkc.edu)
Seth Septer (ssepter@cmh.edu)

Version: 2
Date: 25 November 2014

Author's response to reviews:

- Added the abstract to the second page of the manuscript
- Added the Competing Interests and Author Contributions to the manuscript
- Added line and page numbers to help with the review process
- Deleted extraneous figure